Author’s response to reviews

Title: Perspectives on contraceptive implant use in women living with HIV in Cape Town, South Africa: a qualitative study among primary healthcare providers and stakeholders

Authors:

Anna Brown (annajhbrown@gmail.com)
Jane Harries (jane.harries@uct.ac.za)
Diane Cooper (dcooper@uwc.ac.za)
Chelsea Morroni (chelseaamorroni@gmail.com)

Version: 1 Date: 19 May 2019

Author’s response to reviews:

Please see attached response letter.